# Pricing Powered by Artificial Intelligence: An Assessment Model for the Sustainable Implementation of AI Supported Price Functions

## Anett ERDMANN<sup>1,2</sup>, Morteza YAZDANI<sup>3,\*</sup>, Jose Manuel MAS IGLESIAS<sup>1,2</sup>, Cristina MARIN PALACIOS<sup>1</sup>

<sup>1</sup> ESIC University, Pozuelo de Alarcón, Madrid, Spain

<sup>2</sup> ESIC Business & Marketing School, Pozuelo de Alarcón, Madrid, Spain

<sup>3</sup> SCODeM Research Group, Universidad Internacional de Valencia-VIU, 46002, Valencia, Spain e-mail: anett.erdmann@esic.university, morteza.yazdani@professor.universidadviu.com, josemanuel.mas@esic.university, cristina.marin@esic.university

Received: October 2023; accepted: May 2024

**Abstract.** Artificial Intelligence (AI) in the price management process is being applied in business practice and research to a variety of pricing use cases that can be augmented or automated, providing opportunities as a forecasting tool or for price optimization. However, the complexity of evaluating the technology to prioritize implementation is challenging, especially for small and medium enterprises (SMEs), and guidance is sparse. Which are the relevant stakeholder criteria for a sustainable implementation of AI for pricing purpose? Which type of AI supported price functions meet these criteria best? Theoretically motivated by the hedonic price theory and advances in AI research, we identify nine criteria and eight AI supported price functions (AISPF). A multiple attribute decision model (MADM) using the fuzzy Best Worst Method (BWM) and fuzzy combined compromise solution (CoCoSo) is set up and evaluated by pricing experts from Germany and Spain. To validate our results and model stability, we carried out several random sensitivity analyses based on the weight of criteria exchange. The results suggest accuracy and reliability as the most prominent attribute to evaluate AISPF, while ethical and sustainable criteria are sorted as least important. The AISPF which best meet the criteria are financial prices followed by procurement prices.

**Key words:** price management, artificial intelligence,, human-AI interactions, sustainable AI, multiple attribute decision model.

## 1. Introduction

Price management within the digital environment is increasingly becoming technology based and data driven, with prices being completely or partially set by algorithms (Klein, 2021). In this context, Artificial Intelligence (AI) finds increasing application for sales forecast and price or revenue prediction problems and optimization. To date, a wave of AI point solutions has been proposed (Agrawal *et al.*, 2022), but only few studies provide

<sup>\*</sup>Corresponding author.

a structured framework for AI decisions in marketing (Wu and Monfort, 2022), and to the best of our knowledge, there is no comparative assessment of existing AI support for concrete use cases in price management. Several classification criteria of AI systems have been used (by type of AI, industry usage, mode of human-AI cooperation, primary role of AI, etc.) to investigate which type turns out more attractive in terms of use intention (Kelly et al., 2023; Cabrera-Sánchez et al., 2021; Li et al., 2022; Kim et al., 2021), and the assessment of AI in terms of ethical and sustainable criteria is the focus of the European Commission (2020, 2024), as well as academic research (Dwivedi et al., 2023). However, decision support for investing in such new technologies or data processes is sparse and especially challenging for small and medium-sized enterprises (SME), as managers may not be aware of the ethical and moral implications in a hybrid human-AI decision making process (Sanchez-Hughet et al., 2022; Feuerriegel et al., 2022). AI for process optimization has been studied for industrial processes (Al-Dhaimesh and Taib, 2023), for blockchain networks (Hombalimath et al., 2023), and for financial risk management (Agarwal et al., 2024), all of which highlight the need to explore automation and digitization in an increasingly complex and competitive environment, which can be broken down into several real-world problems to be studied. While research on the criteria and incentives that lead managers to adopt AI in Marketing processes is still sparse, much can be learned from the increasing insights of consumer behaviour towards AI. The use of AI from a consumer perspective is mainly studied building on technology acceptance models and has identified important drivers of people's behaviour intention towards AI (e.g. Kelly et al., 2023; Cabrera-Sánchez et al., 2021; Gursoy et al., 2019). Especially functional AI systems are positively perceived (Kim et al., 2021), and people prefer to collaborate interdependently with AI to reach their goal (Li et al., 2022). The identified behavioural drivers can broadly be classified in economic cost-benefit analysis and social considerations, and to the best of our knowledge, there are so far no studies considering explicitly these criteria in management processes with human-AI interactions.

The objective of the present study is to contribute to the decision making of AI implementation for a responsible and successful pricing process, enhancing the use of AI for SME. For this purpose, theoretically motivated by the hedonic pricing theory (Lancaster, 1966; Rosen, 1974), we focus on the implementation of concrete use cases in the form of AI supported price functions (AISPF). In line with Al-Dhaimesh and Taib (2023), we identify concrete real-world pricing problems where disruptive technologies like AI can improve the task and consequently overcome potential problems or increase competitiveness. Concretely, two main research questions are set up: Which are the most relevant criteria for a sustainable use of AI for pricing? Which existing AI supported pricing functions meet these criteria best?

To address these questions, we make use of multi-criteria techniques for decision making, which find increasing application in the assessment of disruptive technologies (Fallahpour *et al.*, 2021). Concretely, we build on expert opinions setting up a multiple attribute decision model (MADM), using the Best Worst Method by Rezaei (2015) for criteria assessment, and combined compromise solution (CoCoSo) to rate and rank price functions. We identify nine assessment criteria for AISPF, which are differentiated in legal aspects (data protection, non-discrimination, transparency, safety-risk for users) and economic criteria (enhance customer value, accuracy and reliability, cost savings and efficiency, automation, ease of use). For the assessment, eight concrete AISPF are assumed (AI supported procurement prices, AI supported energy prices, AI supported international prices, AI supported personalized pricing, AI supported dynamic pricing, AI supported auction prices, AI supported prices in financial markets, AI supported price index). Pricing experts from Germany and Spain were interviewed and provided their assessment of the defined criteria and the rank of the criteria for each of the AISPF.

The paper contributes to research on AI use for decisions in marketing, especially in sales, and the increasing academic interest in technology based value assessment and pricing, and complements the knowledge on peoples' using intention of AI from a perspective of functional business use. The expert choice, as a tool of active listening of decision makers, is to the best of our knowledge, emloyed for the first time with pricing professionals, as a rising specific profile in sales and marketing. The results provide decision support for firms in implementing and prioritizing AI-supported use cases given the current state of knowledge.

Artificial intelligence is one of the top priorities in both research and business practice. Dwivedi *et al.* (2024) highlights the need to identify stakeholder benefits and tangible measures of the impact of the technology, and to align research and practice, and Agarwal *et al.* (2024) emphasizes the need to prioritize use cases based on different dimensions. Our research contributes to this alignment and addresses the complexity of AI implementation decision making for a specific domain, professional pricing.

However, there are several barriers to delegating pricing decisions to AI, such as risk management framework and governance structure to address the weak accountability of AI (Feuerriegel *et al.*, 2022), with regulatory compliance being one of the main barriers for SMEs (Sanchez-Hughet *et al.*, 2022). In addition, the inability of SMEs to reach the necessary manpower (Sanchez-Hughet *et al.*, 2022) requires a very specific prioritization of the AI implementation in pricing to target the right professionals, not necessarily with highly specialized knowledge in all aspects of pricing, but especially in specific pricing functions that are strategic for the AI implementation.

The aforementioned alignment between research and practice, is particularly critical for Small and Medium-sized Enterprises (SMEs), which often face distinct barriers when integrating AI into their operations. Feuerriegel *et al.* (2022) underscore, for instance, the risk management framework and governance structure to address the weak accountability of AI systems. At the same time, AI for financial tasks and risk management finds plenty of use cases (Agarwal *et al.*, 2024). This challenge, the regulatory compliance in the context of leveraging data through disruptive technologies, has been identified by Sanchez-Hughet *et al.* (2022) as a principal barrier for SMEs. Moreover, the limited access to necessary manpower exacerbates these challenges, making it imperative for SMEs to adopt a highly strategic approach to AI implementation in pricing, to target the right professionals, not necessarily with highly specialized knowledge in all facets of pricing, but especially in specific pricing functions that are strategic for the AI implementation. Such a focused approach ensures that SMEs can navigate the complexities of AI integration, and enhancing their competitive positioning and pricing excellence while aligning the investment with a broad and sustainable stakeholder perspective.

The paper is organized as follows: Section two introduces the challenges SMEs face in implementing AI, stakeholder criteria, and introduces the study case of evaluating AI within the price management process based on identified use cases. Section three outlines the multiple-attribute decision-making approach. Section four presents the results for the ranking of AI pricing functions and the relative importance of the identified criteria. Section five discusses the results and their implications, and concludes the paper.

## 2. Literature Review

The use of technology and data to create value and monetize marketing efforts has been accelerated by the adoption of artificial intelligence in the form of pre-trained, large-scale language models that find many different use cases (Rossi et al., 2024). This increasing importance of big data analytics as competitive advantage has encouraged researchers to provide guidance for SMEs on how to approach the use of big data (Sanchez-Hughet et al., 2022) and prioritize applications (Agarwal et al., 2024). It has been demonstrated that the practice of technological innovation is significantly associated with business performance and SME survival (Rahman and Gogate, 2016), but there are few SMEs that claim to use artificial intelligence and include these technologies as part of their business model (Stentoft et al., 2021). The existence of several barriers to delegating decisions to AI hinders the implementation of, for example, the risk management framework and governance structure to address the weak accountability of AI (Feuerriegel et al., 2022), with regulatory compliance being one of the main barriers for SMEs (Sanchez-Hughet et al., 2022). In addition, the inability of SMEs to reach the necessary manpower (Sanchez-Hughet et al., 2022) requires a very specific prioritization of the AI implementation in processes like pricing to target the right professionals, not necessarily with highly specialized knowledge in all aspects of pricing, but especially in specific pricing functions that are strategic for AI implementation. Hence, it is important to ensure that SMEs can access and use AI (EU Commission, 2020), since they have fewer resources and experience in managing new technologies, however, there is no clear guidance and support. Researchers have studied various determinants of SMEs' technology adoption strategies, nevertheless, there is a small number of published studies which introduces a comprehensive framework to implement these findings (Darban and Wan Ismail, 2012).

To assess the use of AI, ethical and legal aspects need to be taken into account, as established in the whitepaper by the European Commission (2020). This need has also been confirmed by academic research, with managers often being unaware of ethical and moral implications when using hybrid human-AI processes for decision making (Feuerriegel *et al.*, 2022). Moreover, first studies in this line have revealed incentives for unethical practices in human resource management which calls for explicit requirements (Méndez-Suárez *et al.*, 2023), and others have studied AI assessment in this line in financial management or market research (Agarwal *et al.*, 2024; Dwivedi *et al.*, 2023). However, considering concrete marketing functions, an explicit study on the role of these challenges and corresponding decision support has not yet been studied.

With respect to the actions taken by the European Union, a great effort has been made in regulating issues related to data protection in the use of emerging technologies and AI, within the legislative framework set out in the General Data Protection Regulation (GDPR), and this field remains a challenge, especially when most AI users are unaware of and do not exercise control over the use of their data (Wachter and Mittelstadt, 2019) and the role of AI is under attack with customers questioning the abuse of their private data (Mazurek and Małagocka, 2019). As the main raw material of any AI system is data, there is a concern and interest in the use of data, with AI systems having a high capacity for processing large amounts of personal data. It is not just a matter of being aware of whether AI complies with data protection legislation, but of knowing what data is being used and for what purposes (EU Commission, 2020). Authors from academia and industry use this variable in the framework that will help stakeholders evaluate commercial AI solutions (Omoumi *et al.*, 2021).

Another basic aspect about AI is the programming that its algorithms must have so that they do not generate discrimination on the basis of gender, sex, etc. (EU Commission, 2020). Algorithmic decision-making can threaten human rights, such as the right to non-discrimination. We need to bear in mind that legal instruments have severe weaknesses when applied to artificial intelligence. Demetzou (2020) proposes a framework, based on Data Protection Impact Assessment (DPIA) to assess the right to non-discrimination as one of the key fundamental rights that GDPR aims to safeguard.

Furthermore, transparency in the context of AI is an important concern, with the literature often referring to explainability, with reference to both interpretability, as well as trust in the systems (Saura *et al.*, 2022). Transparency has also been increasingly highlighted in regulatory development, company policies, as well as ethical guidelines over the last few years. For example, the EU adopted a strategy on AI in April 2018, and appointed the High-Level Expert Group (AI HLEG) to give advice on both investment strategies, as well as ethical issues with regards to AI in Europe (Larsson and Heintz, 2020). With the use of AI solutions, we cede a certain decision-making capacity to software, derived from its high data processing capacity and machine learning. Therefore, transparency is present in several AI assessment models.

Moreover, risk assessment plays a key role in Safety Management Systems. The introduction of artificial intelligence within autonomous systems makes it hard to reason about the probability and consequences of adverse events when control applications must use previous training sets to guide their response to novel situations (Johnson *et al.*, 2018). Especially when there are repercussions on the life and safety of users it is essential to know how these systems will assume or disregard risks for users (EU Commission, 2020). In this way, the regulation developed by the European Union on Artificial Intelligence establishes four levels of risk. The essence of this rule is to regulate AI according to its capacity to cause harm to society following a "risk-based" approach: the higher the risk, the stricter the rules (EU Commission, 2024).

Considering economic- and management criteria, the use of AI in general allows to improve the value proposition of the firm, through customization, value co-creation, search optimization or optimization of processes (Moreno-Izquierdo *et al.*, 2018), which allows to enhance the customer value, understood from the customer perspective, as well as from the firm perspective on customer value. Furthermore, the accuracy of forecasts is a critical element for decision making of investment and using AI can significantly improve the valuation accuracy, but often needs to be traded off with interpretability and does not always return profitable covering the realized investment (Sakri and Ali, 2022; Liu et al., 2018; Moreno-Izquierdo et al., 2018). The cost saving of predictions using AI algorithms comes especially in terms of time savings in the price prediction, which happens fast or even in real-time (Agrawal et al., 2022). Moreover, the need for automation and scaling is increasingly emphasized in the context of sales and revenue management (Skiera, 2022), with an important aspect of AI technologies being the supervised and unsupervised learning, which allows to create sales artifacts to autonomously perform parts or the whole process (Singh et al., 2018). Finally, considering the insights from consumer research with respect to the acceptance of AI, which has most commonly built on technology acceptance models (in particular TAM and UTAUT) to identify relevant factors, the perceived ease of use and effort expectancy, as well as perceived usefulness, are the most relevant factors inducing a positive behaviour towards AI use (Kelly et al., 2023). The ease of use of AI solutions (or effort expectancy) is also considered a potential barrier for the user intention of AI technologies by managers. However, the results are ambiguous, since recent results on technology adoption of AI based products find that ease of use is not necessarily relevant for adoption (de Blanes Sebastián et al., 2022). This may be explained with the respondents being digital natives or particular user profiles. In this line, first studies provide guidance for segmentation of AI users (Cabrera-Sánchez et al., 2021). Concretely, these authors find that trust in the technology and technology fear moderates the relation between behavioural intention and use behaviour of AI, and hence is considered a crucial aspect in defining user segments which allows to guide investment in particular features, value communication or promoting trust, depending on the target segment, for a successful implementation. Likewise, Kelly et al. (2023) state that age differences in capabilities and difficulties of using AI have been identified by previous research to influence the attitude towards AI (Kelly et al., 2023). Altogether, this effect of personal traits of consumers using AI is expected to be reflected in the user intention by management.

Considering the complexity in the assessment of intelligent technology, several authors have used a differentiation in different blocks of appraisal. For instance, Gursoy *et al.* (2019) recently proposed the AIDUA model (AI Device Use Acceptance Model), which builds on the cognitive appraisal theory. The model assesses consumers' perception in three appraisal stages, namely importance and relevance to themselves in the social context and the cost-benefit analysis, which both determine the outcome in terms of intended behaviour. In the same line, Cabrera-Sánchez *et al.* (2021) highlight two dimensions, the hedonic motivation and expected performance, as the greatest influence on consumers' behavioural intention with respect to AI. Hence, when considering the behaviour of professionals (not consumers) coping with AI technology, in line with the findings from the technology adoption by consumers, we differentiate the identified criteria in (1) social and ethical aspects and (2) economic considerations (Table 1).

Considering the price management process, the digitalization of the sales process implies opportunities and threats for sales professionals, who find themselves in a plurality of roles, often acting like brokers who orchestrate assets and resources, and may have a disruptive impact on the profession in how value is created and functions executed (Singh *et al.* 2019, 2018). Pricing consultants often limit themselves to what can be done in Excel, but pricing in the digital, technological, and socioeconomic context is becoming more and more complex, and the most successful companies are leveraging technology for transformative pricing management (PPS, 2023; Erdmann, 2024). This means that companies are encouraged to make smart pricing investments, and hybrid human-machine solutions offer new opportunities in this transition, which require first to understand how technology contributes to all aspects of pricing, from the cost-perspective to the customer-centric value perspective, and respond to competitors, pursuing opportunities and business growth while minimizing costs.

The hedonic price theory has long been the framework to study price functions as packages of a variety of attributes and relationships between prices (Lancaster, 1966; Rosen, 1974). The use of AI algorithms for price functions based on observed data (attributes, behaviour, external conditions, etc.) changes the approach to price functions, suggesting the pattern instead of imposing a particular (linear) algorithm, but is based on the same input elements as the hedonic pricing models. Several studies have compared the performance of hedonic pricing and AI based methods and find improved accuracy of the value forecast and a reduced price prediction error (Sakri and Ali, 2022; Moreno-Izquierdo *et al.*, 2018; Liu *et al.*, 2018). The trade-off comes often in terms of interpretability, which remains a black-box in many AI algorithms, while being easy for a hedonic approach.

Given the complexity of pricing, academics and pricing professionals differentiate the development of particular stages or the whole process across firms and categories (Simon and Fassnacht, 2019; Frohmann, 2023). Especially at the price implementation stage, AI technology can support a set of different price functions. To date, research has considered different price functions which can be powered by AI, however, there is no joint consideration or comparative assessment respectively in order to provide decision support. We know that the attitude towards AI usage depends on the primary role of the AI system in the context of human-machine interactions, and that functional AI systems are more positively perceived compared to social AI, which can be explained by the mediating effect of perceived usefulness (Kim et al., 2021). Furthermore, the interaction between human and AI can take different forms. Li et al. (2022) define four human AI-cooperation modes, based on the cooperation theory, differentiating between forms of cooperative behaviour (independent versus interdependent) and the goal of the cooperation (exploration versus exploitation/optimization of existing resources). Given the strong and weak aspects of humans and AI, respectively (e.g. large scale tasks versus emotional judgement and random events), the authors identify that people prefer to collaborate interdependently with AI. Concretely, it turns out that people are most positively receptive for human-AI collaboration in a context of interdependent exploitative collaboration. In the context of AI price functions, this implies that a particular AI supported price function may be less or more favoured depending on the type of human-AI interaction.

The main use cases of AI supported pricing can be categorized in (1) forecasting tools in form of AI-based price prediction models, with use cases being financial prices, auction prices, energy prices, procurement prices, exchange rates, or consumer price index and (2) price optimization tools in form of AI-based pricing strategies based on market conditions, with use cases being personalized pricing or segmentation and dynamic pricing.

## AI-Based Price Forecasting Use Cases:

Stock market prices are increasingly being predicted using AI algorithms, encompassing machine learning and deep learning techniques. Ferreira *et al.* (2021) show how market forecasting has shifted from traditional methods to complex AI algorithms that can handle large amounts of data and have demonstrated significant financial gains in stock trading through advanced data analysis and pattern recognition capabilities, suggesting a central role for AI in financial decision-making processes.

Auction prices can also be accurately forecasted using AI, with various algorithms such as genetic algorithms, artificial neural networks, and regression analysis proving to significantly enhance the precision of real estate auction price predictions, as demonstrated by Kang *et al.* (2020). The authors emphasize that this advancement aids investors and fund managers in making more informed decisions, highlighting AI's growing impact on improving economic outcomes in real estate auctions.

Electricity prices are subject to complex and volatile market dynamics, making accurate forecasting essential. Zaroni *et al.* (2020) show how several AI-based models can be used to forecast day-ahead electricity prices in the Finnish market. The complexity and volatility of energy underscores the need of prediction accuracy and the AI models show a superior performance to improve forecasting with great precision.

Procurement prices for agricultural products like strawberries can greatly benefit from AI-based forecasting models. Jafari *et al.* (2020) demonstrate that using daily time-series data on price and yield variables, their AI models can significantly enhance the accuracy of 1- to 7-day ahead price forecasts for strawberries in Canada. This improved forecasting capability offers substantial commercial advantages by providing more accurate market signals and better decision-making tools for procurement processes, thereby optimizing procurement pricing strategies.

International price setting can be significantly enhanced by the application of AI models, as investigated by Chen and Hu (2019). They focus on the exchange rate pass-through (ERPT) effect on Chinese export price indices, demonstrating AI's capability to offer superior forecasting performance. This advancement aids stakeholders in navigating the complexities of international trade and economics with more accurate and informed decisionmaking processes, highlighting the critical role of AI in global pricing mechanisms.

Consumer Price Index (CPI) forecasting has greatly benefitted from advancements in AI, with Ambukege *et al.* (2017) demonstrating a marked improvement in prediction accuracy. Their research integrates neural networks and fuzzy logic to create a potent machine learning model that efficiently utilizes data from the Tanzania National Bureau of Statistics for CPI prediction. This progress not only exemplifies AI's capability in enhancing economic forecasting but also underlines its value in refining decision-making processes, offering a robust tool for strategic planning in various economic sectors. Similar results hold for the AI based prediction of house price index (Xu and Zhang, 2022).

## AI-Based Price Optimization Use Cases:

Personalized pricing, as detailed by Gerlick and Liozu (2020), highlights the pivotal role of AI and algorithmic decision-making in creating price settings that adjust specifically to individual consumer behaviours and preferences. AI based personalized pricing allows to analyse vast consumer data sets, enabling dynamic, real-time price adjustments for

Assessment criteria	AI supported Price Functions (use cases)						
Legal aspects	1. AI supported procurement pricing						
1. Data protection	(Jafari <i>et al.</i> , 2020)						
(EU Commission, 2020, 2024; Wachter and	2. AI supported forecasting of energy prices						
Mittelstadt, 2019; Mazurek and Małagocka, 2019)	(Zaroni <i>et al.</i> , 2020)						
2. Non-discrimination	3. AI supported international pricing						
(EU Commission, 2020, 2024; Demetzou, 2020)	(Chen and Hu, 2019)						
3. Transparency	4. AI supported personalized pricing						
(EU Commission, 2020, 2024; Saura et al., 2022;	(Gerlick and Liozu, 2020)						
Larsson and Heintz, 2020)	5. AI supported dynamic pricing						
4. Safety-risk for users	(Moreno-Izquierdo et al., 2018)						
(EU Commission, 2020, 2024)	6. AI supported prediction for financial markets						
	(Ferreira et al., 2021)						
Economic/Management aspects	7. AI supported auction prices						
5. Enhance customer value	(Kang et al., 2020)						
(Moreno-Izquierdo et al., 2018)	8. AI supported price index						
6. Accuracy and Reliability of the forecast	(Ambukege et al., 2017; Xu and Zhang, 2022)						
(Sakri and Ali, 2022; Liu et al., 2018;	9. AI based ads pricing for web advertising						
Moreno-Izquierdo et al., 2018)	(Park, 2020)						
7. Cost savings and efficiency							
(Agrawal <i>et al.</i> , 2022)							
8. Automation and scaling							
(Skiera, 2022; Singh et al., 2018)							
9. Ease of use (no-code or low-code applications)							
(Kelly et al., 2023; de Blanes Sebastián et al., 2022)							

Table 1 Assessment criteria for AI use and AI supported pricing functions.

a tailored shopping experience. Such advancements in pricing strategy benefit businesses by optimizing profitability and potentially enhance customer satisfaction with prices that accurately reflect each consumer's perceived value. In this context, Gautier *et al.* (2020) studies the technological, economic, and legal perspective on AI's role in personalized pricing, particularly through the lens of algorithmic price discrimination and tacit collusion. The AI algorithms, exploiting massive volumes of personal data, allow for finergrained price discrimination, enhancing the accuracy of personalized pricing strategies. However, first concerns respective potential regulatory needs are raised in this context.

Dynamic pricing in the tourism sector, as studied by Moreno-Izquierdo *et al.* (2018) for Airbnb listings, is significantly enhanced through AI, notably machine learning models. The authors show that AI can analyse extensive databases to optimize pricing strategies effectively. Such data-driven insights enable dynamic adjustment of prices in real-time according to market demand, improving competitiveness and profitability by leveraging sophisticated pattern recognition and data analysis techniques.

Table 1 provides a summary table of the identified use cases, which are denoted as AI supported price functions (AISPF) to be evaluated, and the discussed assessment criteria. The AI algorithms that find applications for the indicated price functions are specified in Table A1 in Appendix A of the paper, with Supervised Machine Learning Algorithms and Reinforcement learning algorithms being the most used.

All these AI-based price functions are promising in improving accuracy at low costs or speed up price adjustments and promotions adapting to customer value perceptions

Authors	AI in marketing	AI in finance	AI in process/task	Pricing automation with AI	Guidance for SMEs	AI assessment using MADM		
Ambukege et al. (2017)		1	1	1	1			
Moreno-Izquierdo <i>et al.</i> (2018)	1		1	1	1			
Ferreira et al. (2021)		1	1	1				
Cabrera-Sánchez <i>et al.</i> (2021)	1		1		1			
Darban and Wan Ismail (2012)	1		1		1			
de Blanes et al. (2022)	1		1		1			
Sanchez-Hughet <i>et al.</i> (2022)	1		1		1			
Gursoy et al. (2019)	1							
Singh <i>et al.</i> (2018)	1		1					
Xu and Zhang (2022)	1			1				
Khan and Nazir (2023)	1					1		
Gerlick and Liozu (2020)			1	1	1			
Demetzou (2020)			1		1			
Feuerriegel et al. (2022)			1		1			
Kim et al. (2021)	1		1					
Li et al. (2022)			1					
Omoumi et al. (2021)			1					
Wachter and Mittelstadt			1					
(2019)								
Jafari <i>et al.</i> (2020)		1	1					
Agrawal et al. (2022)	1		1					
Kelly et al. (2023)	1							
Liu (2024)		1	1			1		

Table 2 Research contribution embedded in the literature.

and market conditions, leading to improved decision making and providing a competitive advantage.

Identifying optimal AI pricing functions, where optimal is defined as meeting the set of identified stakeholder criteria, is relevant for prioritizing investments from a sustainable perspective. That is, while AI-based personalized pricing is highly attractive and beneficial for marketing, care must be taken not to violate privacy or introduce unintended biases (e.g. Rossi *et al.*, 2024). Starting with the implementation of AI in the pricing process, AI point solutions for cost-considerations in the price setting task are identified as a "safe haven", meeting the overall stakeholder criteria, and therefore may be prioritized in getting started with AI in pricing, especially if firms have difficulties to keep up with legal aspects and compliance of new laws such as the AI act and/or are more risk averse in this aspect lacking financial and/or legal support in case of mistakes.

Table 2 highlights the increasing work on the use of AI in task or process automation or augmentation in marketing, as well as in finance for isolated tasks (AI point solutions). This type of complexity in decision making or implementation of disruptive technologies has been addressed in different contexts using multivariate decision models, and the novelty from the methodological approach in this context is the application to AI use for Price setting, as one of the crucial decision variables in the Marketing Mix, considering all encompassing pricing related tasks where AI point solutions have been identified in the literature, which allows to provide guidance to SMEs in AI investment for pricing meeting not only short term needs but overall stakeholder criteria. In addition, the fuzzy approach takes into account the uncertainty of pricing experts in assessing the performance of AI in different pricing functions, since pricing experts are generally not AI experts and the state of AI performance for marketing tasks is currently still a subject of research. The AI implementation in marketing, as well as finance, is rising at an impressive speed, promising a competitive advantage, but at the same time caution is required to implement the technology, and researchers have warned not to rely blindly on pre-trained AI models (Rossi *et al.*, 2024). This requires an assessment of the technology in a broader context, and this paper contributes by evaluating pricing use cases for prioritizing AI investments. For further research, this approach of an assessment of AI point solutions in a particular field using MADM, can be applied to the other variables of the Marketing Mix, as well as finance.

## 3. Methodology

To address the complex decision problem of implementing AI supported price functions, we use multiple criteria decision making (MCDM) approach, in form of multiple-attribute decision making (MADM) approach, which allows to establish rankings of the identified criteria and the assessment of the contribution by alternative AISPF. This methodological approach has been used in a variety of complex decision-making problems associated with economic return. For example, Liu (2024) evaluates the financial risk of high-tech companies in a complex financially volatile environment and managers' uncertainty in decision making using MADM with CoCoSo method illustrated with a case study, emphasizing the widespread attention of this approach in engineering, as well as management science. Rezaei (2015) introduced the BWM to determine weight coefficients of criteria for an MCDM problem through an optimization model and various extensions of the traditional BWM have been developed in recent years. Fuzzy BWM (Guo and Zhao, 2017) is an example. Moreover, BWM attracted a lot of attention from researchers dealing with complicated MCDM problems in different fields and industries such as waste management, supply chain management (Lahri et al., 2021), and transportation management (Rezaei et al., 2018). The algorithm and full description of Fuzzy BWM can be found in Torkayesh et al. (2022) and briefly Fuzzy BWM is explained here:

**Step 1.** First of all, decision criteria are identified. For an MCDM problem, we define several criteria with *n* criteria, and we show them as  $C_1, C_2, \ldots, C_n$ .

**Step 2.** We determine the best criterion (most important) and worst criterion (least important).

**Step 3.** Performing the Fuzzy pairwise comparison for the best criterion. A fuzzy pairwise comparison is produced for all included criteria using the fuzzy linguistic scale (Table 3). A best-to-others vector must be shown as a vector:  $(\tilde{a}_{B1}, \tilde{a}_{B2}, ..., \tilde{a}_{Bn})$  where  $\tilde{a}_{Bj}$  represents the preference of the best criterion over the criterion *j*.

Trinagular fuzzy numbers	Linguistic terms
(1, 1, 1)	Equally important (EI)
(2/3, 1, 3/2)	Weakly important (WI)
(3/2, 2, 5/2)	Fairly important (FI)
(5/2, 3, 7/2)	Very important (VI)
(7/2, 4, 9/2)	Absolutely important (AI)

Table 3 Fuzzy linguistic scale for FBWM.

**Step 4.** Performing the Fuzzy pairwise comparison for the worst criterion. Results are called as the other-to-worst vector. An other-to-worst vector is shown:  $\tilde{A}_W = (\tilde{a}_{1W}, \tilde{a}_{2W}, \dots, \tilde{a}_{nW})$  where  $\tilde{a}_{jW}$  represents the preference of the criterion *j* over the criterion worst criterion *W*.

Step 5. The optimal weightings of criteria can be defined as the formula below:

$$\min \max_{j} \left\{ \left| \frac{\tilde{W}_{B}}{\tilde{W}_{j}} - \tilde{a}_{Bj} \right|, \left| \frac{\tilde{W}_{j}}{\tilde{W}_{W}} - \tilde{a}_{jW} \right| \right\}$$
(1)

$$\sum_{j} R(\tilde{W})_{j} = 1,$$
  

$$\alpha_{j}^{w} < \beta_{j}^{w} < \gamma_{j}^{w},$$
  

$$l_{i}^{w} \ge 0 \quad \text{for all } j$$

where  $\tilde{W}_B = (\alpha_B^w, \beta_B^w, \gamma_B^w)$ ,  $\tilde{W}_j = (\alpha_j^w, \beta_j^w, \gamma_j^w)$ ,  $\tilde{W}_W = (\alpha_W^w, \beta_W^w, \gamma_W^w)$ ,  $\tilde{a}_{Bj} = (\alpha_{Bj}^w, \beta_{Bj}^w, \gamma_{Bj}^w)$ , and  $\tilde{a}_{JW} = (\alpha_{jW}^w, \beta_{jW}^w, \gamma_{jW}^w)$ 

The other method for producing the alternatives score is CoCoSo or combined compromise solution (Yazdani et al., 2019). This method has been investigated in many research applications and its algorithm is different from BWM. Various type of fuzzy CoCoSo are being used as Ecer and Pamucar (2020) use it in supplier selection. However, the classic fuzzy model is used in this study and Peng and Huang (2020) developed fuzzy version of CoCoSo. CoCoSo includes a powerful novel approach to handling complicated MCDA problems. The algorithm is based on an integrated exponential weighted product and several combined strategies. It can be used as a distance measure based on the gray relational coefficient that makes it more flexible. The CoCoSo approach introduces three aggregation strategies (scores) to obtain more confidential outcomes. As a consequence, each score enhances the decision quality and finally reveals a full ranking index (Fernàndez-Portillo et al., 2023; Yazdani et al., 2019). The method increases the decision-making quality in big decision problems (when a high amount of alternatives exist) due to its simplicity and accuracy. In this paper, we apply a fuzzy extension of CoCoSo to deal with uncertainty in supply chain risk. The entire process of formulating a ranking of AI price functions can be observed in Fig. 1. It starts with a deep literature review and expert interview to encounter the relevant alternatives and criteria. Then a decision analysis platform, combined



Fig. 1. The procedure for decision making and pricing function system.

TFN	Linguistic term
(0.222, 0.25, 0.286)	Absolutely less significant (ALS)
(0.25, 0.286, 0.333)	Dominantly less significant (DLS)
(0.286, 0.333, 0.4)	Much less significant (MLS)
(0.333, 0.4, 0.5)	Really less significant (RLS)
(0.4, 0.5, 0.667)	Less significant (LS)
(0.5, 0.667, 1)	Moderately less significant (MoLS)
(0.667, 1, 1)	Weakly less significant (WLS)

Table 4 Linguistic scale for ranking evaluation.

Source: Demir et al. (2022).

by Fuzzy BWM and Fuzzy CoCoSo, will address the best alternative with respect to fundamental objectives (criteria). Finally, a sensitivity analysis of the results and reporting is carried out.

The F-CoCoSo method is applied based on the following steps.

**Step 1.** Identifying the decision-making matrix including criteria, alternatives, decision-making team, questionnaire preparation, etc.

**Step 2.** Experts will offer the opinion for evaluation. They are asked to form a decision matrix using a fuzzy linguistic scale (Table 4) according to:

$$\tilde{X}_{ij} = \begin{bmatrix} x_{11} & \cdots & x_{1n} \\ \vdots & \ddots & \vdots \\ x_{m1} & \cdots & x_{mn} \end{bmatrix}, \quad \text{for } i = 1, \dots, m \text{ and } j = 1, \dots, n.$$
(2)

**Step 3.** The normalized matrix is developed based on the nature of the criteria as equations (3)–(4).

$$\tilde{r}_{ij} = \frac{x_{ij} - \min_i x_{ij}}{\max_i x_{ij} - \min_i x_{ij}},\tag{3}$$

$$\tilde{r}_{ij} = \frac{\max_i x_{ij} - x_{ij}}{\max_i x_{ij} - \min_i x_{ij}},\tag{4}$$

where equation (3) is used for benefit criteria, and equation (4) is used for cost criteria.

**Step 4.** The sum of the weighted comparability sequence  $(SW_i)$  and the power-weighted comparability sequences  $(PW_i)$  for each alternative are determined based on equations (4)–(5).

$$\widetilde{SW}_i = \sum_{j=1}^n (\tilde{w}_j \tilde{r}_{ij}),\tag{5}$$

$$\widetilde{PW}_i = \sum_{j=1}^n (w_j)^{r_{ij}}.$$
(6)

**Step 5.** Then the aggregated appraisal scores are applied to calculate the relative weights of alternatives.

$$\tilde{Q}_{1} = \frac{\widetilde{PW}_{i} + \widetilde{SW}_{i}}{\sum_{i=1}^{m} (PW_{i} + \widetilde{SW}_{i})},$$

$$\widetilde{SW}_{i} = \widetilde{PW}_{i}$$
(7)

$$\tilde{Q}_2 = \frac{SW_i}{\min_i S\widetilde{W}_i} + \frac{PW_i}{\min_i \widetilde{PW}_i},\tag{8}$$

$$\tilde{Q}_3 = \frac{\lambda \left(SW_i\right) + (1-\lambda)(PW_i)}{\lambda \max_i \widetilde{SW}_i + (1-\lambda)\max_i \widetilde{PW}_i},\tag{9}$$

where  $0 \le \lambda \le 1$  and is usually considered 0.5 ( $\lambda = 0.5$  is taken in this study).

**Step 6.** Ultimately, the integrated value for each alternative is obtained based on equation (9).

$$\tilde{Q}_{i} = (\tilde{Q}_{1} \times \tilde{Q}_{2} \times \tilde{Q}_{3})^{\frac{1}{3}} + \frac{1}{3} \left( \tilde{Q}_{1} + \tilde{Q}_{2} + \tilde{Q}_{3} \right).$$
(10)

542

#### 4. Use Cases, Data and Analysis

Our paper considers a specific professional area to study existing AI point solutions and provide an assessment considering stakeholder criteria, which are ranked and evaluated for each AI pricing point solution, taking into account the uncertainty in the assessment by experts. Price managers, at the intersection of marketing and finance, are core to monetizing the firm's efforts, and managers are under pressure to react quickly, foresee changes in any type of costs, and predict consumer perception and price reactions for a large number of stock keeping units and consumer segments. In this context, AI offers a competitive advantage in several dimensions, and there are several AI point solutions, but there is little evidence on which criteria matter for implementation and which AI-based pricing functions meet these criteria best to be prioritized. That is, instead of directly implementing AI in pricing at any point with promising short-term benefits, it is considered crucial first to assess which are the relevant criteria for all stakeholders (firms, consumers, government) and assess the importance for pricing professionals when considering AI use for particular price functions, as well as which of the AI price functions (point solutions) meet the overall stakeholder criteria best.

Our approach is illustrated with a detailed case study on pricing use cases (AI pricing functions), although it's worth noting that our methodology is adaptable to other marketing mix variables or broader business areas. With AI in processes being one of the rising research fields (e.g. Liu, 2024; Hombalimath *et al.*, 2023; Al-Dhaimesh and Taib, 2023), multivariate decision-making models have proven very useful in assessing disruptive technologies to improve decision-making (Khan and Nazir, 2023; Liu, 2024).

We contacted eight international pricing experts from Germany and Spain for their participation in the study, which is a usual size for expert sample (e.g. Azadeh *et al.* (2015) use six experts to assess the lean performance of organizations using AHP and fuzzy scales; Suner *et al.* (2012) use an AHP setting with five experts to determine priorities of criteria for a decision support on cancer treatment methods). The experts received a questionnaire asking them to rate criteria and alternatives. The expert profiles are summarized in Table 5.

The computation is divided in two stages, one is to deliver the criteria weight (the importance factor for each of the considered drivers) by fuzzy BWM and lingo program, and second is to obtain each alternative score using fuzzy CoCoSo software. BWM results are provided in Table 6, which indicates that C6 and C5 are the best criteria. The consistency of the computation has been checked and seems placed in an acceptable area. For the next stage, it is necessary to utilize those weights for CoCoSo as inputs. The algorithm for Fuzzy CoCoSo (F-CoCoSo) can be found in Peng and Huang (2020). After finalizing the computation, Table 7 is generated, showing the alternative score and each alternative ranking. Higher score addresses the better alternative and A6 and A1 are categorized as 1st and 2nd ranked options, while A4 shows the worst alternative. The order of alternatives is as follows:

$$A6 > A1 > A2 > A3 > A5 > A8 > A7 > A4.$$

The computation detail for Fuzzy CoCoSo can be seen in Appendix A, in the Tables A2–A5. In the CoCoSo method, the value of  $\lambda$  is a measure to check the sensitivity

#### A. Erdmann et al.

Table 5
Selection of experts.

Expert	Professional profile
Expert I	Consultant for Pricing and Digital Business Transformation, with 30 years of experience at an
(Germany)	international price consultancy company, as well as freelance. Book author.
	56 years old, economist.
Expert II	Advisor for Pricing and Revenue Management at an international management consultancy
(Germany)	firm, with 15 years of experience. Book author and conference speaker.
	PhD in Operation Management (Revenue Management).
Expert III	Academic in the field of value assessment and pricing, with 14 years of experience in pricing
(Germany)	research, conferences in the field and teaching of price management at an Executive MBA
	programme and a university. 40 years old, PhD in Economics.
Expert IV	Professor, Advisor and Speaker on Marketing, in particular consumer behaviour and pricing.
(Germany)	Director of a management research centre, collaboration with an international pricing
	consultancy. Author of academic articles and pricing case studies. PhD in Business Studies.
Expert V	Revenue and Pricing Manager in an international hotel group, especially dynamic pricing and
(Spain)	revenue management IT systems. MBS and EMBA.
Expert VI	Managing Director at an international telecommunication company, specialized in customer
(Spain)	value and revenue management.
Expert VII	Academic in AI applications in different sector. Lecturer in national and international
(Spain)	universities. 39 years old. PhD in Science and Technology applied to Industrial Engineering.
Expert VIII	Pricing practitioner and CEO of SME technology company. Graduated in marketing and
(Spain)	lecturer of price management courses since 1992.

Table 6
The weight produced by Fuzzy BWM for 9 criteria based on expert opinion.

Criteria	Ex. 1	Ex. 2	Ex. 3	Ex. 4	Ex. 5	Ex. 6	Ex. 7	Ex. 8	Av. Weights
C1	0.110	0.070	0.081	0.149	0.039	0.134	0.066	0.183	0.1040
C2	0.075	0.027	0.047	0.059	0.048	0.037	0.033	0.028	0.0443
C3	0.054	0.093	0.081	0.089	0.086	0.027	0.098	0.042	0.0713
C4	0.026	0.055	0.032	0.119	0.033	0.134	0.049	0.042	0.0612
C5	0.110	0.147	0.142	0.102	0.214	0.214	0.197	0.156	0.1602
C6	0.128	0.253	0.142	0.268	0.179	0.160	0.230	0.226	0.1982
C7	0.154	0.059	0.133	0.095	0.146	0.134	0.131	0.156	0.1260
C8	0.214	0.097	0.133	0.089	0.113	0.107	0.098	0.056	0.1136
C9	0.128	0.199	0.208	0.030	0.142	0.053	0.098	0.110	0.1210

of the results, so based on Table A6 in Appendix A, we can see that while changing  $\lambda$ , the ranking will not alter. Another point is the ranking and alternative score in three different ranking strategies in equations (7), (8) and (9). Once again, our results show the high stability comparing to final ranking by equation (10). In total, the fuzzy CoCoSo method validates the stable and confidential assessment.

In addition, to validate the accuracy of the results and observe how sensitive is the Co-CoSo algorithm in this study, we performed twenty-one sensitivity analysis tests. Tests 1 to 21 include the criteria weight replacement one by one randomly. For example, among them, T2 and T21 are the weight replacement of *"ease of use"* criterion with highest (best) and lowest importance (worst) criteria, respectively. With each test, CoCoSo ranking will change and then ranking comparison allows us to examine the stability and sensitivity of the results. The extracted ranking results are observed in Table 8. Except for the tests 3,

	to F-CoCoSo method	l.
	Alt. score	Ranking
A1	3.6811	2
A2	3.6507	3
A3	3.5936	4
A4	2.9159	8
A5	3.4842	5
A6	3.7062	1
A7	3.2474	7
A8	3.4742	6

Table 7 AI pricing function score and their ranking according to F-CoCoSo method.

Table 8 Sensitivity analysis test and alternative rankings.

	T1	T2	Т3	T4	T5	T6	T7	T8	Т9	T10	T11	T12	T13	T14	T15	T16	T17	T18	T19	T20	T21
A1	2	2	1	2	4	2	2	2	2	2	2	4	4	2	2	2	3	4	3	2	3
A2	3	3	3	3	2	3	3	3	3	3	1	3	2	3	3	3	1	2	2	3	2
A3	4	4	4	4	3	4	4	4	4	4	4	2	3	4	4	4	4	3	4	4	4
A4	8	8	8	8	8	8	8	8	8	8	8	8	8	8	8	8	8	8	8	8	8
A5	6	6	5	6	6	6	5	5	5	6	6	6	6	6	6	5	6	6	6	6	6
A6	1	1	2	1	1	1	1	1	1	1	3	1	1	1	1	1	2	1	1	1	1
A7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7	7
A8	5	5	6	5	5	5	6	6	6	5	5	5	5	5	5	6	5	5	5	5	5

11 and 17, where A6 obtained 2nd and 3rd ranking, the rest of the experiments reflected that A6 is the best alternative. The worst alternative (A4) maintained its position in all the experiments. Moreover, alternatives 5, 7 and 8, except for slight changes, remained within the ranking position. This confirms that our model and obtained results are stable enough and experts can rely on them. Based on the Table 8, we have produced Fig. 2 that shows the ranking stability schematically.

## 5. Discussion, Management Implications and Conclusion

### 5.1. Discussion of the Results

The assessment model reveals that accuracy and reliability of the forecast (C6), followed by enhancement of customer value (C5) and cost saving and efficiency (C7) are the most relevant criteria for the evaluation of AI in price management. That is, in light of unpredictable surrounding external economic conditions and the critical impact of pricing and revenue management on a firm's economic growth, accuracy and reliability emerge as crucial evaluation metrics. Moreover, accuracy often needs to be traded off with interpretability (Liu *et al.*, 2018) and profitability of the technical investment (Moreno-Izquierdo *et al.*, 2018), which emphasizes the need to focus on this criterion. Likewise, the trend of a value-based approach to price setting and the monetization of this value (Hinterhuber and Liozu, 2019) justifies the importance of the effect on customer value. Finally, in third place, a technological innovation in the management process should not only reduce costs



Fig. 2. The random sensitivity analysis experiments for alternatives.

but also be efficient, which confirms this aspect being considered one of the main benefits of AI in a different context (Agrawal *et al.*, 2022).

The four least important criteria are all within the same category of ethical and sustainable aspects. Concretely, non-discrimination (C2) ranks as least relevant, and slightly higher are safety risk (C4), transparency (C3) and data protection (C1). Note that all these criteria are discussed in the whitepaper by the European Union (2020), which confirms the existence of market failure in terms of ethical and sustainable implementation of AI, and emphasizes the need for regulatory guidelines.

Finally, the results reveal that the AI supported price functions which rank highest in the contribution to the established set of stakeholder criteria are the prediction of financial prices (A6), followed by procurement prices (A1). Especially AI supported financial price prediction is already finding broad application in the business practice (Ferreira *et al.*, 2021). On the opposite side, AI supported personalized pricing (A8) ranks lowest in the expected contribution to the established criteria. The latter is the result of the low importance that is given to C1–C4, which mainly aim to protect customers. Our research result was tested through several sensitivity analysis for approval and accuracy.

However, some results are inconsistent, since some experts evaluate ease of use as very important while for others it is one of the least important criteria. We argue that this may depend on the level of management as pricing professionals where the expert operates or on his or her technical knowledge. This is in line with recent articles of technology adoption of AI, which do not identify ease of use as a relevant factor (de Blanes Sebastián *et al.*, 2022), which may be due to certain cohorts being used for digital operations. Hence, this gives rise to further research, segmenting the assessment of pricing professionals by age or the executive level of the expert. In any case, this limitation of the study is in line with Singh *et al.* (2018), who find that practitioners and scholars differ in their prediction on how much digitalization technologies like AI will cause disruptions and how

salespeople and AI will work together. Note that this is in agreement with the segmentation of consumers towards their AI user intention, suggested by Cabrera-Sánchez *et al.* (2021).

#### 5.2. Management Implications

The increasing interest in generative AI among leading tech giants highlights its perceived economic and strategic importance. However, this enthusiasm also raises ethical concerns, emphasizing the need for a balanced approach to AI implementation in price management. As Kotler (2024) asserts, the strategic application of AI in marketing decisions is crucial for company success in the coming years. Research by Hagendorff *et al.* (2023) and Jansen *et al.* (2023) demonstrates how AI can outperform traditional problem-solving tasks in certain contexts, suggesting its potential for enhancing decision-making processes.

These aspects have also been highlighted in the context of AI-based pricing management, as shown by the results of the evaluation by professional pricing consultants, with the most important evaluation criteria by the professionals being superior performance in terms of accuracy, value and cost. However, the prevailing focus on immediate business outcomes (short-term perspective) has so far given little attention to the need for research on AI that promotes a sustainable (long-term) perspective, which requires consideration of a broader stakeholder perspective.

In this regard, our study underscores the importance of integrating professional expertise and judgment with respect to AI use cases with ethical and legal considerations, to advance the future of price management. AI models are particularly noted for their role in predictive analysis, and their potential to be cheaper and easier to use compared to pure human analysis, offering an alternative or complement to conventional methods. Despite these advances, the ethical implications of AI in decision-making require a careful in-depth examination (Dwivedi *et al.*, 2023). That is, when using AI for pricing, the low ranking of ethical aspects for certain pricing functions requires special care in implementing AI for these tasks, building on the AI model as co-intelligence rather than pure automation to achieve intelligent pricing solutions while respecting ethical requirements.

Echoing the call from McKinsey of the need to prioritize AI use cases in finance (Agarwal *et al.*, 2024), we highlight the critical need for a holistic assessment of AI use cases in pricing – an interdisciplinary field between marketing and finance – to ensure a comprehensive understanding of their impacts. This approach allows for an informed prioritization of AI applications by SMEs, taking into account not just economic benefits but also ethical and legal considerations.

The need for ethical and human criteria is less relevant for AI-based price forecasting use cases, but more relevant for AI use cases that consider price optimization, such as personalized pricing and dynamic pricing. This requires special care for the latter, and the need to train and hire professionals to work with AI, with particular attention to data use, transparency, and security threats, as well as algorithmic biases that may lead to unethical discrimination. That is, in line with Agarwal *et al.* (2024) for financial risk management and Dwivedi *et al.*, our contribution is to provide a detailed assessment of the complex context of using AI in price management, balancing technological innovation that promises economic gain with ethical responsibility.

## 5.3. Conclusion

The present research has provided first directions in the assessment and implementation of concrete AI solutions for pricing, provides evidence of market failure for a sustainable implementation of AI in price management and opens directions for further research. The study brings a combination of MCDM methods and expert opinions to address required solution for decision problems on the use of AI technology, which is a novel approach in this context and can be recommended to researchers to advance their findings. In addition, utilizing other fuzzy set models, MCDM methods like MABAC (Multi-Attributive Border Approximation Area Comparison), MARCOS (Measurement of alternatives and ranking according to COmpromise solution), EDAS (Evaluation based on Distance from Average Solution) and stratified decision-making tools can enhance the accuracy of the results in decision analysis for the implementation of disruptive technologies in marketing processes, as they are part of future research lines.

## A. Appendix

AI algorithms are classified into three types, (1) Supervised Machine Learning Algorithm (SMLA), which identifies the relationship between input variables and output variables (Regression, chatbot). (2) Unsupervised Learning Algorithm (USLA), which has no output variables and all variables play the same role (Principal Component Analysis (PCA)). (3) Reinforcement learning algorithm (RLA), which makes decisions based on experience and learning from mistakes (Artificial neural network (ANN), neuro fuzzy system (NFS)). Table 2 provides an overview of the dominant algorithm for each AISPF.

AI supported Price Functions (AISPP) to be evaluated	AI Algorithm
1. AI supported procurement pricing	SMLA as a Regression (Jafari <i>et al.</i> , 2020) and as a Chatbot (Cui <i>et al.</i> , 2022)
2. AI supported forecasting of	RLA as an ANN (Alshater et al., 2022; El Hag Hassan and
energy prices	El-Khawaga, 2024; Kumari and Tanwar, 2021; Jasiński, 2020)
3. AI supported international pricing	RLA as an ANN (Safari and Ghavifekr, 2021) and as NFS (Petković et al., 2020)
4. AI supported personalized pricing	RLA as an ANN (Zhang et al., 2020; Häckel et al., 2022)
5. AI supported dynamic pricing	RLA (Tiwari <i>et al.</i> , 2021) with geolocations (Pandey and Caliskan, 2021) or Blockchain (Tiwari <i>et al.</i> , 2021)
6. AI supported prediction for	USLA as PCA (Yu et al., 2021)
financial markets	SMLA (Regression) (Le Gallo, 2021)
	RLA (Antunes et al., 2021)
7. AI supported auction prices	RLA as an ANN (Munro and Rassenti, 2019; Ghosh et al., 2022)
8. AI supported price index	SMLA as a Regression) (Jafari <i>et al.</i> , 2020) as a chatbot (Cui <i>et al.</i> , 2022)
9. AI based ads pricing for web advertising	RLA as a NFS (Sarkar and Pal, 2021) and ANN (Modarresi and Diner, 2019)

Table A1 SML and RLA as frequently used AI algorithms in AISPF.

Table A2 Initial fuzzy decision matrix.

	C1	C1 C2				C3			C4			C5			C6			C7			C8			C9			
A1	5.00	7.00	8.63	3.75	5.75	7.63	3.75	5.75	7.50	5.75	7.75	9.25	2.75	4.50	6.38	6.00	8.00	9.50	6.00	8.00	9.50	5.75	7.75	9.25	4.50	6.50	8.38
A2	5.50	7.50	9.00	4.75	6.75	8.38	5.25	7.25	8.75	6.00	8.00	9.38	2.13	3.75	5.75	5.75	7.75	9.13	5.50	7.50	9.00	6.00	8.00	9.50	5.00	7.00	8.75
A3	5.75	7.75	9.13	5.00	7.00	8.50	4.25	6.25	7.88	5.75	7.75	9.13	4.25	6.25	8.25	4.75	6.75	8.63	4.50	6.50	8.25	5.25	7.25	8.88	4.25	6.25	8.13
A4	1.88	3.75	5.75	2.25	4.25	6.25	1.63	3.50	5.50	4.25	6.25	8.13	6.00	8.00	9.38	6.25	8.25	9.63	3.50	5.50	7.25	4.50	6.50	8.38	3.25	5.25	7.25
A5	2.75	4.75	6.75	2.63	4.50	6.50	2.25	4.00	6.00	4.75	6.75	8.63	5.25	7.25	8.88	6.00	8.00	9.50	4.75	6.75	8.38	5.25	7.25	9.00	3.50	5.50	7.50
A6	5.25	7.25	8.88	4.25	6.25	8.13	5.50	7.50	9.13	4.75	6.75	8.50	3.88	5.75	7.50	5.50	7.50	9.13	5.00	7.00	8.75	5.75	7.75	9.25	4.75	6.75	8.50
A7	4.00	6.00	7.88	2.63	4.75	6.25	3.25	5.25	7.13	4.75	6.75	8.63	5.50	7.50	9.25	4.50	6.50	8.38	3.50	5.50	7.38	4.50	6.50	8.38	4.25	6.25	8.13
<b>A8</b>	5.25	7.25	8.88	5.00	7.00	8.63	5.50	7.50	9.13	5.25	7.25	9.00	3.25	5.00	6.75	5.50	7.50	9.25	3.00	4.75	6.50	5.25	7.25	9.00	4.75	6.75	8.50

Table A3 Normalized fuzzy matrix.

	C1			C2			C3			C4			C5			C6			C7			C8			C9		
A1	0.43	0.71	0.93	0.24	0.55	0.84	0.28	0.55	0.78	0.29	0.68	0.98	0.09	0.33	0.59	0.29	0.68	0.98	0.46	0.77	1.00	0.25	0.65	0.95	0.23	0.59	0.93
A2	0.50	0.78	0.98	0.39	0.71	0.96	0.48	0.75	0.95	0.34	0.73	1.00	0.00	0.22	0.50	0.24	0.63	0.90	0.38	0.69	0.92	0.30	0.70	1.00	0.32	0.68	1.00
A3	0.53	0.81	1.00	0.43	0.75	0.98	0.35	0.62	0.83	0.29	0.68	0.95	0.29	0.57	0.84	0.05	0.44	0.80	0.23	0.54	0.81	0.15	0.55	0.88	0.18	0.55	0.89
A4	0.00	0.26	0.53	0.00	0.31	0.63	0.00	0.25	0.52	0.00	0.39	0.76	0.53	0.81	1.00	0.34	0.73	1.00	0.08	0.38	0.65	0.00	0.40	0.78	0.00	0.36	0.73
A5	0.12	0.40	0.67	0.06	0.35	0.67	0.08	0.32	0.58	0.10	0.49	0.85	0.43	0.71	0.93	0.29	0.68	0.98	0.27	0.58	0.83	0.15	0.55	0.90	0.05	0.41	0.77
A6	0.47	0.74	0.97	0.31	0.63	0.92	0.52	0.78	1.00	0.10	0.49	0.83	0.24	0.50	0.74	0.20	0.59	0.90	0.31	0.62	0.88	0.25	0.65	0.95	0.27	0.64	0.95
A7	0.29	0.57	0.83	0.06	0.39	0.63	0.22	0.48	0.73	0.10	0.49	0.85	0.47	0.74	0.98	0.00	0.39	0.76	0.08	0.38	0.67	0.00	0.40	0.78	0.18	0.55	0.89
<b>A8</b>	0.47	0.74	0.97	0.43	0.75	1.00	0.52	0.78	1.00	0.20	0.59	0.93	0.16	0.40	0.64	0.20	0.59	0.93	0.00	0.27	0.54	0.15	0.55	0.90	0.27	0.64	0.95

Table A4 Fuzzy weighted comparability.

	C1			C2			C3			C4			C5			C6			C7			C8			C9			Si		
A1	0.04	0.07	0.10	0.01	0.02	0.04	0.02	0.04	0.06	0.02	0.04	0.06	0.01	0.05	0.09	0.06	0.14	0.19	0.06	0.10	0.13	0.03	0.07	0.11	0.03	0.07	0.11	0.28	0.61	0.88
A2	0.05	0.08	0.10	0.02	0.03	0.04	0.03	0.05	0.07	0.02	0.04	0.06	0.00	0.04	0.08	0.05	0.13	0.18	0.05	0.09	0.12	0.03	0.08	0.11	0.04	0.08	0.12	0.29	0.62	0.88
A3	0.06	0.08	0.10	0.02	0.03	0.04	0.02	0.04	0.06	0.02	0.04	0.06	0.05	0.09	0.14	0.01	0.09	0.16	0.03	0.07	0.10	0.02	0.06	0.10	0.02	0.07	0.11	0.24	0.58	0.87
A4	0.00	0.03	0.06	0.00	0.01	0.03	0.00	0.02	0.04	0.00	0.02	0.05	0.09	0.13	0.16	0.07	0.15	0.20	0.01	0.05	0.08	0.00	0.05	0.09	0.00	0.04	0.09	0.16	0.50	0.78
A5	0.01	0.04	0.07	0.00	0.02	0.03	0.01	0.02	0.04	0.01	0.03	0.05	0.07	0.11	0.15	0.06	0.14	0.19	0.03	0.07	0.10	0.02	0.06	0.10	0.01	0.05	0.09	0.21	0.54	0.84
A6	0.05	0.08	0.10	0.01	0.03	0.04	0.04	0.06	0.07	0.01	0.03	0.05	0.04	0.08	0.12	0.04	0.12	0.18	0.04	0.08	0.11	0.03	0.07	0.11	0.03	0.08	0.12	0.28	0.62	0.90
A7	0.03	0.06	0.09	0.00	0.02	0.03	0.02	0.03	0.05	0.01	0.03	0.05	0.07	0.12	0.16	0.00	0.08	0.15	0.01	0.05	0.08	0.00	0.05	0.09	0.02	0.07	0.11	0.16	0.50	0.81
<b>A8</b>	0.05	0.08	0.10	0.02	0.03	0.04	0.04	0.06	0.07	0.01	0.04	0.06	0.02	0.06	0.10	0.04	0.12	0.18	0.00	0.03	0.07	0.02	0.06	0.10	0.03	0.08	0.12	0.23	0.55	0.84

A. Erdmann et al.

 Table A5

 Fuzzy exponentially weighted comparability sequence.

	C1			C2			C3			C4			C5			C6			C7			C8			C9			Pi		
A1	0.92	0.96	0.99	0.94	0.97	0.99	0.91	0.96	0.98	0.93	0.98	1.00	0.68	0.84	0.92	0.78	0.93	1.00	0.91	0.97	1.00	0.85	0.95	0.99	0.84	0.94	0.99	7.75	8.49	8.87
A2	0.93	0.97	1.00	0.96	0.98	1.00	0.95	0.98	1.00	0.94	0.98	1.00	0.00	0.79	0.89	0.76	0.91	0.98	0.89	0.95	0.99	0.87	0.96	1.00	0.87	0.95	1.00	7.16	8.49	8.86
A3	0.94	0.98	1.00	0.96	0.99	1.00	0.93	0.97	0.99	0.93	0.98	1.00	0.82	0.91	0.97	0.55	0.85	0.96	0.83	0.92	0.97	0.81	0.93	0.98	0.81	0.93	0.99	7.58	8.46	8.86
A4	0.00	0.87	0.94	0.00	0.95	0.98	0.00	0.91	0.95	0.00	0.94	0.98	0.90	0.97	1.00	0.81	0.94	1.00	0.72	0.89	0.95	0.00	0.90	0.97	0.00	0.88	0.96	2.44	8.25	8.73
A5	0.80	0.91	0.96	0.88	0.95	0.98	0.84	0.92	0.96	0.87	0.96	0.99	0.87	0.95	0.99	0.78	0.93	1.00	0.85	0.93	0.98	0.81	0.93	0.99	0.69	0.90	0.97	7.39	8.38	8.81
A6	0.92	0.97	1.00	0.95	0.98	1.00	0.95	0.98	1.00	0.87	0.96	0.99	0.80	0.89	0.95	0.72	0.90	0.98	0.86	0.94	0.98	0.85	0.95	0.99	0.85	0.95	0.99	7.79	8.52	8.89
A7	0.88	0.94	0.98	0.88	0.96	0.98	0.90	0.95	0.98	0.87	0.96	0.99	0.88	0.95	1.00	0.00	0.83	0.95	0.72	0.89	0.95	0.00	0.90	0.97	0.81	0.93	0.99	5.95	8.31	8.78
<b>A8</b>	0.92	0.97	1.00	0.96	0.99	1.00	0.95	0.98	1.00	0.90	0.97	1.00	0.74	0.86	0.93	0.72	0.90	0.99	0.00	0.85	0.92	0.81	0.93	0.99	0.85	0.95	0.99	6.87	8.40	8.81

								e							
	Fuzz	y Q1		Crisp Fuzzy Q2				Crisp	Fuzz	y Q3		Crisp	Final	Qi	Final
				Q1				Q2				Q3	Score Qi		ranking
A1	0.10	0.13	0.18	0.14	4.92	7.27	9.13	7.11	0.82	0.93	1.00	0.92	3.68	3.6811	2
A2	0.10	0.13	0.18	0.13	4.77	7.35	9.13	7.08	0.76	0.93	1.00	0.90	3.65	3.6507	3
A3	0.10	0.13	0.18	0.13	4.62	7.06	9.04	6.91	0.80	0.92	0.99	0.91	3.59	3.5936	4
A4	0.03	0.12	0.17	0.11	2.01	6.47	8.46	5.65	0.27	0.89	0.97	0.71	2.92	2.9159	8
A5	0.10	0.12	0.18	0.13	4.34	6.81	8.81	6.66	0.78	0.91	0.99	0.89	3.48	3.4842	5
A6	0.10	0.13	0.18	0.14	4.95	7.32	9.22	7.17	0.82	0.93	1.00	0.92	3.71	3.7062	1
A7	0.08	0.12	0.17	0.13	3.44	6.50	8.62	6.19	0.62	0.90	0.98	0.83	3.25	3.2474	7
A8	0.09	0.12	0.18	0.13	4.25	6.90	8.87	6.67	0.73	0.91	0.99	0.88	3.47	3.4742	6

Table A6 Final ranking of alternatives.

#### References

- Agarwal, R., Kremer, A., Kristensen, I., Luget, A. (2024). How generative AI can help banks manage risk and compliance. McKinsey. https://www.mckinsey.com/capabilities/risk-and-resilience/our-insights/howgenerative-ai-can-help-banks-manage-risk-and-compliance.
- Agrawal, A., Gans, J., Goldfarb, A. (2022). Power and Prediction: The Disruptive Economics of Artificial Intelligence. HBR Press, Boston.
- Al-Dhaimesh, S.H., Taib, N. (2023). A review: investigation of augmented reality–BIM benefits in design process in AEC industry. *Informatica*, 47(5). https://doi.org/10.31449/inf.v47i5.4671.
- Alshater, M.M., Kampouris, I., Marashdeh, H., Atayah, O.F., Banna, H. (2022). Early warning system to predict energy prices: the role of artificial intelligence and machine learning. *Annals of Operations Research*, 1(37). https://doi.org/10.1007/s10479-022-04908-9.
- Ambukege, G., Justo, G., Mushi, J. (2017). Neuro fuzzy modelling for prediction of consumer price index. Journal of Artificial Intelligence Research, 8(5), 33–44. https://doi.org/10.48550/arXiv.1710.05944.
- Antunes, M., Santiago, A.R., Manso, S., Regateiro, D., Barraca, J.P., Gomes, D., Aguiar, R.L. (2021). Building an IoT platform based on service containerisation. *Sensors*, 21(19), 6688. https://doi.org/10.3390/s21196688.

Azadeh, A., Zarrin, M., Abdollahi, M., Noury, S., Farahmand, S. (2015). Leanness assessment and optimization by fuzzy cognitive map and multivariate analysis. *Expert Systems with Applications*, 42(15–16), 6050–6064.

- Cabrera-Sánchez, J.P., Villarejo-Ramos, Á.F., Liébana-Cabanillas, F., Shaikh, A.A. (2021). Identifying relevant segments of AI applications adopters–expanding the UTAUT2's variables. *Telematics and Informatics*, 58, 101529. https://doi.org/10.1016/j.tele.2020.101529.
- Chen, H., Hu, W. (2019). Research on exchange rate pass-through effect based on artificial intelligence approach. Concurrency and Computation: Practice and Experience, 31(9), e4986. https://doi.org/10.1002/cpe.4986.
- Cui, T., Suleman, M.T., Zhang, H. (2022). Do the green bonds overreact to the COVID-19 pandemic? *Finance Research Letters*, 49, 103095. https://doi.org/10.1016/j.frl.2022.103095.
- Darban, M., Wan Ismail, W.K. (2012). Advanced manufacturing technology adoption in SMEs: an integrative model. *Journal of Technology Management & Innovation*, 7(4), 112–120. https://doi.org/10.4067/S0718-27242012000400009.
- de Blanes Sebastián, M.G., Sarmiento-Gueda, J.R., Antonovica, A. (2022). Application and extension of the UTAUT2 model for determining behavioral intention factors in use of the artificial intelligence virtual assistance. *Frontiers in Psychology*, 13. https://doi.org/10.3389/fpsyg.2022.993935.
- Demetzou, K. (2020). Processing operations 'likely to result in a high risk to the rights and freedoms of natural persons'. In: Antunes, L., Naldi, M., Italiano, G., Rannenberg, K., Drogkaris, P. (Eds.), *Privacy Technologies* and Policy, APF 2020, Lecture Notes in Computer Science, Vol. 12121. Springer, Cham. https://doi.org/ 10.1007/978-3-030-55196-4\_2.
- Demir, G., Damjanović, M., Matović, B., Vujadinović, R. (2022). Toward sustainable urban mobility by using fuzzy-FUCOM and fuzzy-CoCoSo methods: the case of the SUMP podgorica. *Sustainability*, 14(9), 4972.
- Dwivedi, Y.K., Kshetri, N., Hughes, L., Slade, E.L., Jeyaraj, A., Kar, A.K., Baabdullah, A.M., Koohang, A., Raghavan, V., Ahuja, M., Albanna, H., Albashrawi, M.A., Al-Busaidi, A.S., Balakrishnan, J., Barlette, Y., Basu, S., Bose, I., Brooks, L., Buhalis, D., Wright, R. (2023). So what if ChatGPT wrote it? Multidisciplinary

perspectives on opportunities, challenges and implications of generative conversational AI for research, practice and policy. *International Journal of Information Management*, 71, 102642.

- Dwivedi, Y.K., Jeyaraj, A., Hughes, L., Davies, G.H., Ahuja, M., Albashrawi, M.A., Walton, P. (2024). "Real impact": challenges and opportunities in bridging the gap between research and practice – making a difference in industry, policy, and society. *International Journal of Information Management*, 78 102750.
- Ecer, F., Pamucar, D. (2020). Sustainable supplier selection: a novel integrated fuzzy best worst method (F-BWM) and fuzzy CoCoSo with Bonferroni (CoCoSo'B) multi-criteria model. *Journal of Cleaner Production*, 266, 121981. https://doi.org/10.1016/j.jclepro.2020.121981.
- El Hag Hassan, K., El-Khawaga, S. (2024). Comparative study of rice straw recycling systems. Journal of Sustainable Agricultural and Environmental Sciences, 3(1), 18–27. https://doi.org/10.21608/jsaes. arXiv:2024.252709.1067.
- Erdmann, A. (2024). Pricing 360°: a paradigm shift in the way pricing is managed in the digital economy. In: *The Impact of Digitalization on Current Marketing Strategies*. Emerald Publishing Limited, pp. 79–96.
- EU Commission (2020). White Paper on Artificial Intelligence: A European approach to excellence and trust. European Commission, Brussels.
- EU Commission (2024). Shaping Europe's digital future: AI Act. https://bit.ly/3T1O3KL.
- Fallahpour, A., Yazdani, M., Mohammed, A., Wong, K.Y. (2021). Green sourcing in the era of industry 4.0: towards green and digitalized competitive advantages. *Industrial Management & Data Systems*, 121(9), 1997–2025. https://doi.org/10.1108/IMDS-06-2020-0343.
- Fernàndez-Portillo, L.A., Yazdani, M., Estepa-Mohedano, L., Sisto, R. (2023). Prioritisation of strategies for the adoption of organic agriculture using BWM and fuzzy CoCoSo. Soft Computing, 1–32.
- Ferreira, F.G.D.C., Gandomi, A.H., Cardoso, R.T.N. (2021). Artificial intelligence applied to stock market trading: a review. *IEEE Access*, 9, 30898–30917. https://doi.org/10.1109/ACCESS.2021.3058133.
- Feuerriegel, S., Shrestha, Y.R., von Krogh, G., Zhang, C. (2022). Bringing artificial intelligence to business management. *Nature Machine Intelligence*, 4(7), 611–613. https://doi.org/10.1038/s42256-022-00512-5.
- Frohmann, F. (2023). Digital Pricing: A Guide to Strategic Pricing for the Digital Economy. Springer.
- Gautier, A., Ittoo, A., Van Cleynenbreugel, P. (2020). AI algorithms, price discrimination and collusion: a technological, economic and legal perspective. *European Journal of Law and Economics*, 50(3), 405–435. https://doi.org/10.1007/s10657-020-09662-6.
- Gerlick, J.A., Liozu, S.M. (2020). Ethical and legal considerations of artificial intelligence and algorithmic decision-making in personalized pricing. *Journal of Revenue and Pricing Management*, 19, 85–98. https:// doi.org/10.1057/s41272-019-00225-2.
- Ghosh, I., Chaudhuri, T.D., Alfaro-Cortés, E., Gámez, M., García, N. (2022). A hybrid approach to forecasting futures prices with simultaneous consideration of optimality in ensemble feature selection and advanced artificial intelligence. *Technological Forecasting and Social Change*, 181, 121757. https://doi.org/10.1016/ j.techfore.2022.121757.
- Guo, S., Zhao, H. (2017). Fuzzy best-worst multi-criteria decision-making method and its applications. *Knowledge-Based Systems*, 121, 23–31. https://doi.org/10.1016/j.knosys.2017.01.010.
- Gursoy, D., Chi, O.H., Lu, L., Nunkoo, R. (2019). Consumers acceptance of artificially intelligent (AI) device use in service delivery. *International Journal of Information Management*, 49, 157–169. https://doi.org/10.1016/ j.ijinfomgt.2019.03.008.
- Häckel, B., Karnebogen, P., Ritter, C. (2022). AI-based industrial full-service offerings: a model for payment structure selection considering predictive power. *Decision Support Systems*, 152, 113653. https://doi.org/10. 1016/j.dss.2021.113653.
- Hagendorff, T., Fabi, S., Kosinski, M. (2023). Human-like intuitive behavior and reasoning biases emerged in large language models but disappeared in ChatGPT. *Nature Computational Science*, 3(10), 833–838.
- Hinterhuber, A., Liozu, S.M. (2019). Pricing Strategy Implementation: Translating Pricing Strategy into Results. Routledge.
- Hombalimath, A., Mangla, N., Balodi, A. (2023). Designing a permissioned blockchain network for the insurance claim process using hyperledger fabric and composer. *Informatica*, 47(3). https://doi.org/10.31449/ inf.v47i3.4158.
- Jafari, F., Mousavi, J., Ponnabalam, K., Karray, F., Nassar, L. (2020). Machine learning tools for the prediction of fresh produce procurement price. In: 2020 IEEE International Conference on Systems, Man, and Cybernetics (SMC), pp. 3091–3095. https://doi.org/10.1109/SMC42975.2020.9282887.
- Jansen, T., Heitmann, M., Reisenbichler, M., Schweidel, D.A. (2023). Automated Alignment: Guiding Visual Generative AI for Brand Building and Customer Engagement. Available at SSRN 4656622.

- Jasiński, A. (2020). Public space or safe space-remarks during the COVID-19 pandemic. *Technical Transactions*, 117(1). https://doi.org/10.37705/TechTrans/e2020020.
- Johnson, K.W., Torres Soto, J., Glicksberg, B.S., Shameer, K., Miotto, R., Ali, M., Dudley, J.T. (2018). Artificial intelligence in cardiology. *Journal of the American College of Cardiology*, 71(23), 2668–2679. https://doi.org/10.1016/j.jacc.2018.03.521.
- Kang, Z., Zhou, W., Zhao, Z., Shao, J., Han, M., Xu, Z. (2020). Large-scale multi-view subspace clustering in linear time. *Proceedings of the AAAI Conference on Artificial Intelligence*, 344, 4412–4419. https://doi.org/ 10.1609/aaai.v34i04.5867.
- Kelly, S., Kaye, S.A., Oviedo-Trespalacios, O. (2023). What factors contribute to acceptance of artificial intelligence? A systematic review. *Telematics and Informatics*, 77, 101925. https://doi.org/10.1016/j.tele.2022. 101925.
- Khan, H.U., Nazir, S. (2023). Assessing the role of AI-based smart sensors in smart cities using AHP and MOORA. Sensors, 23(1), 494. https://doi.org/10.3390/s23010494.
- Kim, J., Merrill, K. Jr., Collins, K.C. (2021). AI as a friend or assistant: the mediating role of perceived usefulness in social AI vs. functional AI. *Telematics and Informatics*, 64, 101694. https://doi.org/10.1016/j.tele. 2021.101694.
- Klein, T. (2021). Autonomous algorithmic collusion: Q-learning under sequential pricing. RAND Journal of Economics, 52(3), 538–558. https://doi.org/10.1111/1756-2171.12383.
- Kotler, P.A. (2024). Lifetime in marketing: lessons learned and the way ahead. Journal of Marketing.
- Kumari, A., Tanwar, S. (2021). A reinforcement-learning-based secure demand response scheme for smart grid system. *IEEE Internet of Things Journal*, 9(3), 2180–2191.
- Lahri, V., Shaw, K., Ishizaka, A. (2021). Sustainable supply chain network design problem: using the integrated BWM, TOPSIS, possibilistic programming, and ε-constrained methods. *Expert Systems with Applications*, 168, 114373. https://doi.org/10.1016/j.eswa.2020.114373.
- Lancaster, K.J. (1966). A new approach to consumer theory. *Journal of Political Economy*, 74, 132–157. https:// doi.org/10.1086/259131.
- Larsson, S., Heintz, F. (2020). Transparency in artificial intelligence. *Internet Policy Review*, 9(2). https://doi.org/ 10.14763/2020.2.1469.
- Le Gallo, J. (2021). Cross-section spatial regression models. In: *Handbook of Regional Science*. Berlin, Heidelberg, Springer Berlin Heidelberg, pp. 2117–2139. https://doi.org/10.1007/978-3-662-60723-7\_8.
- Li, J., Huang, J., Liu, J., Zheng, T. (2022). Human-AI cooperation: modes and their effects on attitudes. *Telem*atics and Informatics, 73, 101862. https://doi.org/10.1016/j.tele.2022.101862.
- Liu, H. (2024). Enhanced CoCoSo method for intuitionistic fuzzy MAGDM and application to financial risk evaluation of high-tech enterprises. *Informatica*, 48(5). https://doi.org/10.31449/inf.v48i5.5169.
- Liu, Z., Yan, S., Cao, J., Jin, T., Tang, T., Yang, J., Wang, O. (2018). A Bayesian approach to residential property valuation based on build environment and house characteristics. In: *IEEE International Conference on Big Data*, pp. 1455–1464.
- Mazurek, G., Małagocka, K. (2019). Perception of privacy and data protection in the context of the development of artificial intelligence. *Journal of Management Analytics*, 6(4), 344–364. https://doi.org/10.1080/ 23270012.2019.1671243.
- Méndez-Suárez, M., de Obesso, M.D.L.M., Márquez, O.C., Palacios, C.M. (2023). Why do companies employ prohibited unethical artificial intelligence practices? In: *IEEE Transactions on Engineering Management*. https://doi.org/10.1109/TEM.2023.3258686.
- Modarresi, K., Diner, J. (2019). An evaluation metric for content providing models, recommendation systems, and online campaigns. In: *Computational Science – ICCS 2019: 19th International Conference*, Faro, Portugal, June 12–14, 2019, Proceedings, Part II, Vol. 19. Springer International Publishing, pp. 550–563. https://doi.org/10.1007/978-3-030-22741-8\_39.
- Moreno-Izquierdo, L., Egorova, G., Peretó-Rovira, A., Más-Ferrando, A. (2018). Exploring the use of artificial intelligence in price maximisation in the tourism sector: its application in the case of Airbnb in the Valencian Community. *Journal of Regional Research*, 42, 113–128. http://hdl.handle.net/10045/86772.
- Munro, D.R., Rassenti, S.J. (2019). Combinatorial clock auctions: price direction and performance. Games and Economic Behavior, 117, 195–217. https://doi.org/10.1016/j.geb.2019.07.001.
- Omoumi, P., Ducarouge, A., Tournier, A., Harvey, H., Kahn, C.E., Louvet-de Verchère, F., Richiardi, J. (2021). To buy or not to buy—evaluating commercial AI solutions in radiology (the ECLAIR guidelines). *European Radiology*, 31, 3786–3796. https://doi.org/10.1007/s00330-020-07684-x.

- Pandey, A., Caliskan, A. (2021). Disparate impact of artificial intelligence bias in ridehailing economy's price discrimination algorithms. In: *Proceedings of the 2021 AAAI/ACM Conference on AI, Ethics, and Society*, pp. 822–833. https://doi.org/10.1145/3461702.3462561.
- Park, H. (2020). MLP modeling for search advertising price prediction. Journal of Ambient Intelligence and Humanized Computing, 11(1), 411–417.
- Peng, X., Huang, H. (2020). Fuzzy decision making method based on CoCoSo with critic for financial risk evaluation. *Technological and Economic Development of Economy*, 26(4), 695–724. https://doi.org/10.3846/ tede.2020.11920.
- Petković, B., Petković, D., Kuzman, B., Milovančević, M., Wakil, K., Ho, L.S., Jermsittiparsert, K. (2020). Neuro-fuzzy estimation of reference crop evapotranspiration by neuro fuzzy logic based on weather conditions. *Computers and Electronics in Agriculture*, 173, 105358. https://doi.org/10.1016/j.compag.2020. 105358.
- Rahman, T., Gogate, V. (2016). Learning ensembles of cutset networks. Proceedings of the AAAI Conference on Artificial Intelligence, 30(1), 3301–3307. https://doi.org/10.1609/aaai.v30i1.10428.
- Rezaei, J. (2015). Best-worst multi-criteria decision-making method. Omega, 53, 49–57. https://doi.org/10.1016/ j.omega.2014.11.009.
- Rezaei, J., van Roekel, W.S., Tavasszy, L. (2018). Measuring the relative importance of the logistics performance index indicators using Best Worst Method. *Transport Policy*, 68, 158–169. https://doi.org/10.1016/ j.tranpol.2018.05.007.
- Rosen, S. (1974). Hedonic prices and implicit markets: product differentiation in pure competition. *Journal of Political Economy*, 82(1), 35–55.
- Rossi, S., Rossi, M., Mukkamala, R.R., Thatcher, J.B., Dwivedi, Y.K. (2024). Augmenting research methods with foundation models and generative AI. *International Journal of Information Management*, 77 102749.
- Safari, A., Ghavifekr, A.A. (2021). International stock index prediction using artificial neural network (ANN) and Python programming. In: 2021 7th International Conference on Control, Instrumentation and Automation (ICCIA). IEEE, pp. 1–7.
- Sakri, S.B., Ali, Z. (2022). Analysis of the dimensionality issues in house price forecasting modeling. In: 2022 Fifth International Conference of Woman in Data Science at Price Sultan University, pp. 13–19. https:// doi.org/10.1109/WiDS-PSU54548.2022.00015.
- Sanchez-Hughet, C., Aramendia-Muneta, M.E., Erro-Garces, A. (2022). Seizing opportunities in Europe: a roadmap for efficient big data implementation in Spanish SMEs. *Digital Policy, Regulation and Governance*, 24(5), 463–478. https://doi.org/10.1108/DPRG-02-2022-0019.
- Sarkar, A., Pal, B. (2021). Competitive pricing strategies of multi channel supply chain under direct servicing by the manufacturer. *RAIRO-Operations Research*, 55, S1849–S1873. https://doi.org/10.1051/ro/2020063.
- Saura, J.R., Ribeiro-Soriano, D., Palacios-Marqués, D. (2022). Assessing behavioral data science privacy issues in government artificial intelligence deployment. *Government Information Quarterly*, 39(4), 101679.
- Simon, H., Fassnacht, M. (2019). Price Management: Strategy, Analysis, Decision, Implementation. Springer, Cham, Switzerland.
- Singh, J., Flaherty, K., Sohi, R.S., Deeter-Schmelz, D., Habel, J., Meunier-FitzHugh, K.L., Malshe, A., Mullins, R., Onyemah, V. (2018). Sales profession and professionals in the age of digitization and artificial intelligence technologies: concepts, priorities, and questions. *Journal of Personal Selling & Sales Management*, 39(1), 2–22. https://doi.org/10.1080/08853134.2018.1557525.
- Skiera, B. (2022). Challenges of marketing automation: linking MarTech & SalesTech. NIM Artificial Intelligence Review, 14(2), 11–17.
- Stentoft, J., Adsbøll Wickstrøm K, K., Philipsen, K., Haug, A. (2021). Drivers and barriers for Industry 4.0 readiness and practice: empirical evidence from small and medium-sized manufacturers. *Production Planning & Control*, 32(10), 811–828.
- Suner, A., Çelikoğlu, C.C., Dicle, O., Sökmen, S. (2012). Sequential decision tree using the analytic hierarchy process for decision support in rectal cancer. *Artificial Intelligence in Medicine*, 56(1), 59–68. https://doi.org/ 10.1016/j.artmed.2012.05.003.
- Tiwari, A.K., Abakah, E.J.A., Le, T.L., Leyva-de la Hiz, D.I. (2021). Markov-switching dependence between artificial intelligence and carbon price: the role of policy uncertainty in the era of the 4th industrial revolution and the effect of COVID-19 pandemic. *Technological Forecasting and Social Change*, 163, 120434. https://doi.org/10.1016/j.techfore.2020.120434.
- Torkayesh, A.E., Yazdani, M., Ribeiro-Soriano, D. (2022). Analysis of industry 4.0 implementation in mobility sector: an integrated approach based on QFD, BWM, and stratified combined compromise solution un-

der fuzzy environment. Journal of Industrial Information Integration, 30, 100406. https://doi.org/10.1016/ j.jii.2022.100406.

- Wachter, S., Mittelstadt, B. (2019). A right to reasonable inferences: re-thinking data protection law in the age of big data and AI. *Columbia Business Law Review*, 2. https://doi.org/10.1145/3287560.3287574.
- Wu, C.-W., Monfort, A. (2022). Role of artificial intelligence in marketing strategies and performance. *Psychology and Marketing*, 40(3), 484–496. https://doi.org/10.1002/mar.21737.
- Xu, X., Zhang, Y. (2022). Second-hand house price index forecasting with neural networks. *Journal of Property Research*, 39(3), 215–236. https://doi.org/10.1080/09599916.2021.1996446.
- Yazdani, M., Zarate, P., Zavadskas, E.K., Turskis, Z. (2019). A combined compromise solution (CoCoSo) method for multi-criteria decision-making problems. *Management Decision*, 57(9), 2501–2519. https:// doi.org/10.1108/MD-05-2017-0458.
- Yu, L., Mei, Q., Xiang, L., Liu, W., Mohamad, N.I., István, B., Fernandez, J., Gu, Y. (2021). Principal component analysis of the running ground reaction forces with different speeds. *Frontiers in Bioengineering and Biotechnology*, 9, 629809. https://doi.org/10.3389/fbioe.2021.629809.
- Zhang, S., Xiang, C., Xu, S. (2020). 6G: connecting everything by 1000 times price reduction. *IEEE Open Journal of Vehicular Technology*, 1, 107–115.
- Zaroni, A., Piazzi, A., Tettamanti, T., Sleisz, A. (2020). Investigation of day-ahead price forecasting models in the finnish electricity market. In: *Proceedings of the 12th International Conference on Agents and Artificial Intelligence (ICAART 2020)*, pp. 829–835.

**A. Erdmann** is a professor and academic director of the Marketing Department at ESIC University and director of the University Master in Marketing Management (MUDM) at ESIC University. She holds a PhD in economics and a master's degree in economic analysis from the Universidad Carlos III de Madrid and is a fellow of the HEA. Her research focuses on the value assessment and pricing of consumer technologies and the optimization of digital marketing, leveraging data and technology.

**M. Yazdani** is a professor & senior researcher at Universidad Internacional de Valencia and chief research fellow at Vilnius Tech, Institute of Dynamic Management, Lithuania. He was a visiting professor at University of Foggia and Toulouse University last year. He previously worked at Universidad Loyola Andalucia. Morteza was an associate researcher for RUC-APS project at University of Toulouse and IRIT institute. Morteza handles research in management and operations, specifically decision making theories, supply chain management and sustainable development. He is AE of several journals indexed in web of sciences. **J.M. Mas Iglesias** is a vice chancellor of Academic Policy and Faculty at ESIC University, responsible for the design and implementation of degrees, as well as the faculty. He holds a PhD in media research from UC3M. He is principal investigator of the research group Companies, Institutions and Consumers in the environment of Digital Marketing and Technology Research Group- CICDMT- ESIC-1-M-2020 of ESIC University. He has extensive experience as a professor of strategy and communication and digital marketing in undergraduate, graduate and corporate training. He has more than 15 years of professional experience in the world of communication and advertising, where he has been general manager of agencies at national and international level.

**C. Marin Palacios** ia a professor and academic director of the Department of Computer Science at ESIC University. She holds a degree in mathematical sciences and a PhD in business economics. She is a fellow of the HEA. Her line of research focuses on the application of mathematical models and AI to the social sciences and education.